Author's response to reviews

Title: Combined effect of CO2 enrichment and foliar application of Salicylic acid on anthocyanin, flavonoids and isoflavonoids production and their antioxidant activity in ginger

Authors:

Ali Ghasemzadeh (upmali@yahoo.com)
Hawa ZE Jaafar (Hawaze@gmail.com)
Ehsan Karimi (ehsanb_karimi@yahoo.com)
Ibrahim Mohd Hafiz (mhfizhd@yahoo.com)

Version: 8 Date: 22 September 2012

Author's response to reviews: see over
Dear Dr. Tom Rowles
Executive Editor
BMC Complementary and Alternative Medicine

Thank you very much for your e-mail. We did corrections according to the reviewer comments. Our correction for submitted manuscript as follow:

Title: Combined effect of CO₂ enrichment and foliar application of Salicylic acid on anthocyanin, flavonoids and isoflavonoids production and their antioxidant activity in ginger

Ali Ghasemzadeh*, Hawa Z.E. Jaafar*, Ehsan Karimi, Mohd Hafiz Ibrahim

Reviewer 1: No correction

Reviewer 2 (2806795557994105_comment.pdf)

<table>
<thead>
<tr>
<th>NO</th>
<th>Comments</th>
<th>Corrections</th>
</tr>
</thead>
</table>
| 1  | ● Page No.2 Ref. Arts et al., 2003 or 2002 as in manuscript Ref.  
● Page No.3 Ref. Mattson et al., 2003 or 2005 as in manuscript Ref.  
● Page No.6 Ref. Mensor et al. (2001). Not found in manuscript Ref.  
● Page No.12 Ref. Matros et al., 2006 Not found in manuscript Ref. and Ozeki et al. (2003) or 2006 as in manuscript Ref.  
● Page No.16 Ref. (WANG et al., 2003). Change to Wange et al., 2003 | Corrected |
| 2  | Also, there are some Ref. in Ref. list but did not sited in text | Corrected and all of references were checked again. |

Best regards

Ali Ghasemzadeh, Ph.D  
Department of Crop Science  
Faculty of Agriculture  
University Putra Malaysia  
43400, Serdang, Malaysia  
Tel: +6 0-17-291-7679  
Fax: +60-3-8943 5973